Author's response to reviews

Title: A systematic review of mental disorder, suicide, and deliberate self harm in lesbian, gay and bisexual people

Authors:

Michael King (m.king@medsch.ucl.ac.uk)
Joanna Semlyen (josemlyen@hotmail.com)
Sharon See Tai See Tai (s.seetai@pcps.ucl.ac.uk)
Helen Killaspy (h.killaspy@medsch.ucl.ac.uk)
David Osborn (d.osborn@medsch.ucl.ac.uk)
Dmitri Popelyuk (d_popelyuk@hotmail.com)
Irwin Nazareth (i.nazareth@pcps.ucl.ac.uk)

Version: 9 Date: 8 August 2008

Author's response to reviews:

typos and references corrected for final time - i.e. proofed